Author’s response to reviews

Title: Listening to women's voices: the experience of giving birth with paramedic care in Queensland, Australia

Authors:

Belinda Flanagan (BFlanaga@usc.edu.au)
Bill Lord (wlord@uc.edu.au)
Rachel Reed (rreed1@usc.edu.au)
Gail Crimmins (gcrimmin@usc.edu.au)

Version: 2 Date: 14 Jul 2019

Author’s response to reviews:

Response to reviewer 2 final comment.

Thank you for addressing most of my comments. No response has been given to my comment on the inclusion of women in the sample that were fearful about adverse outcomes or who had a bad experience during the home birth. Although one of the suggested references (Erlandsson 2015) was included, the results are not discussed in view of this reference and why the authors did not find that women were worried about adverse outcomes as in this paper. After addressing these issues I think the paper is suitable for publication.

Response – this has been addressed in paragraph 1 page 17. A further comment has been added.

Note there are no new comments from reviewer 2 to be addressed.